Author’s response to reviews

Title: A Qualitative study of older adults’ perspectives on initiating exercise and mindfulness practice

Authors:
Diana C. Parra (parrad@wustl.edu)
Julie Loebach Wetherell (jwetherell@ucsd.edu)
Alexandria Van Zandt (aevanzandt@wustl.edu)
Ross Brownson (rbrownson@wustl.edu)
Janardan Abhishek (ajanardan@wustl.edu)
Eric Lenze (lenzee@wustl.edu)

Version: 2 Date: 30 Nov 2019

Author’s response to reviews:

We have included the titles/legends of the figures after the references and remove them within the text.